Author's response to reviews

Title: Moderate sensitivity and high specificity of emergency department administrative data for transient ischemic attacks

Authors:

Amy Y. X. Yu (amy.yu@ucalgary.ca)
Hude Quan (hquan@ucalgary.ca)
Andrew McRae (amcrane@ucalgary.ca)
Gabrielle Wagner (gowagner@ucalgary.ca)
Michael Hill (michael.hill@ucalgary.ca)
Shelagh Coutts (scoutts@ucalgary.ca)

Version: 2 Date: 08 Sep 2017

Author’s response to reviews:

We have addressed all the editorial revisions requested and trust our formatting is consistent with BMC Health Services Research editorial policies.